% SIMIODE-TeacherStudent-Template.tex and simiode.cls for LaTeX2e
% Version 22 November 2016.
% 
% This template is to be used with the  class file "simiode.cls". 
%  
% The latest version of this class and template files 
% can be found at
% 
%       <http://www.SIMIODE.org> in the Author section.
%
% A "Quick Start Guide" appears below.
%
% This file must be processed with version LaTeX2e or higher.
% Appreciation to Mark Frantz, IUPUI; John Thoo, Yuba College; and Eric Sullivan, Carroll College.
%
\documentclass{simiode}
%
% With this version of the tex file and the associated SIMIODE.cls file you can produce BOTH a Student Version and a Teacher Version of your modeling scenario or technique narrative.
%
% Student Version contains the title STUDENT VERSION along with the NAME OF SCENARIO, with no abstract, no keywords, no tags, no author identifiers, and no COMMENTS section. The Student Version contains only the STATEMENT of the modeling scenario only with appropriate figures, data, instructions, and perhaps a bibliography appropriate for student resourcing, but nothing in the bibliography which would "give away" the nature of the modeling scenario, etc.
%
%The Teacher Version contains the material in the Student Version, namely the title NAME OF SCENARIO, now headed with   TEACHER VERSION, but also full author  identification, abstract, key words, tag words, and, following the STATEMENT, a section of COMMENTS which can be discussion of presentation  ideas, "solutions," techniques, etc.
%
% Authors can prepare a full Teacher Version using the notions immediately below  to create a pdf for review, but in review please eliminate all information about the author as we use a double-blind referee system. Such information will be restored when accepted and published in SIMIODE at www.simiode.org.
%
%***********   Toggle the following line to turn the teacher version on (1) or off (0). ***********
%
\def\TeacherEdition{0} % Toggle this line's value.
\newcommand{\teacher}[1]{\ifnum\TeacherEdition=1 #1 \fi}
%
% The \teacher{ } command wraps any text that will appear only in the teacher version of
% the document. See this file for examples. This has been done for you in the document below.   
% Rename the pdf after you compile each final version.
%
\usepackage{amsmath,amssymb}
\usepackage{color}
\usepackage{graphicx}
\usepackage{hyperref}
\usepackage{url}
\usepackage{framed}
%
\renewcommand{\baselinestretch}{1.3}
%
% NOTE regarding macros %%
%
% Any macros defined for your paper should be contained in 
% the top matter. Likewise, any environment definitions such 
% as \newtheorem or \newenvironment should also go in the 
% top matter. 
% 
% Do not \input your macros as separate files within your 
% final version because more files make it harder for the 
% editors and users to keep track of and/or modify and 
% append your paper.
%
% Top matter is above and the beginning of the document below.
%
\begin{document}
\begin{thebibliography}{98}


\bibitem{FastestPitches} Baseball-Almanac.com. {\it The Fastest Pitcher in Baseball History\/}. \href{https://www.baseball-almanac.com/articles/fastest-pitcher-in-baseball.shtml}{https://www.baseball-almanac.com/articles/fastest-pitcher-in-baseball.shtml}. Accessed 6 January 2022. 

\bibitem{Dataset}  {\it Diamond Kinetics Pitchtracker, Rapsodo, and Pitch Tracking Technologies\/}. \href{https://www.drivelinebaseball.com/2018/07/diamond-kinetics-pitchtracker-rapsodo-pitch-tracking-technologies}{https://www.drivelinebaseball.com/2018/07/diamond-kinetics-pitchtracker-rapsodo-pitch-\\ tracking-technologies}. Data set at \href{https://docs.google.com/spreadsheets/d/1csJmCHzopbc\_EWa1idFGMoDUBorkBRIbZvatA2nk11U/edit\#gid=2134910657}{https://docs.google.com/spreadsheets/d/1csJmCHzopbc\_E\\Wa1idFGMoDUBorkBRIbZvatA2nk11U/edit\#gid=2134910657}.  Accessed 24 July 2019.

\bibitem{FastBall2015} {\it Fastball\/}. 2015. \href{https://www.youtube.com/watch?v=SIhQlAass2Y}{https://www.youtube.com/watch?v=SIhQlAass2Y}. Accesses 22 January 2022.



\bibitem{StatcastVelocity} MLB.com {\it Velocity (VELO)\/} \href{https://www.mlb.com/glossary/statcast/velocity}{https://www.mlb.com/glossary/statcast/velocity}. Accessed 31 December 2021.

\bibitem{NolanRyan} Wikipedia. {\it Nolan Ryan\/}. \href{https://en.wikipedia.org/wiki/Nolan\_Ryan}{https://en.wikipedia.org/wiki/Nolan\_Ryan}.  Accessed 6 January 2022.



\bibitem{Pitchfx} Wikipedia. {\it PITCHf/x\/}. \href{https://en.wikipedia.org/wiki/PITCHf/x}{https://en.wikipedia.org/wiki/PITCHf/x}. Accessed 6 January 2022.

\bibitem{TrackMan} Wikipedia. {\it TrackMan \/}. \href{https://en.wikipedia.org/wiki/TrackMan}{https://en.wikipedia.org/wiki/TrackMan}. Accessed 6 January 2022.

\end{thebibliography}


\newpage
%
% You only need to change the NAME OF SCENARIO in the next line and delete the footnote. 
% See above for instructions on toggling from Teacher Version to Student Version.
%
\title{\ifnum\TeacherEdition=1 TEACHER VERSION \else STUDENT VERSION \fi\\Fastest Fastball?}




\markboth{Fastest Fastball?}{Fastest Fastball?}

\author{Erich McAlister \\
Department of Mathematics\\
Fort Lewis College, Durango CO 81301 USA\\
mcalister\_e@fortlewis.edu
}

\makeStitlePDFLaTex


\begin{abstract}
We consider the problem of comparing pitch velocities using measurement methods in different eras of baseball.
\end{abstract}



\section*{INTRODUCTION AND SCENARIO DESCRIPTION}
Pitch velocity is one of the most fascinating statistics in baseball, as documented in the 2015 documentary {\it Fastball}   \cite{FastBall2015}. Modern measurements of pitch velocity are taken as the maximum velocity achieved at any point between the pitcher's hand and home plate \cite{StatcastVelocity}. However, the velocity of the ball slows down due to air resistance. Thus, we may assume this is just the instantaneous velocity when the ball leaves the pitcher's hand. In the past the velocity measurements were taken at specific distances from home plate. From 2006 to 2016, pitch velocity was recorded by the PITCHf/x system which measured velocity at a distance of 50.5 feet from home plate (10 feet from the mound) \cite{Pitchfx} \cite{TrackMan}.



\begin{framed}
\noindent {\bf Task 1:} Assume that the velocity, $v$, of a pitched baseball satisfies the differential equation
\[
mv' = KA^a\rho^b v^c, 
\]
where $m$ is the mass of the baseball, $A$ is the cross sectional area of the baseball, $\rho$ is the air density, and $K$ is dimensionless. Show that in order for this model to be dimensionally consistent, we must have
\[
v' = -kv^2
\] 
for some constant $k$. Also, explain why it make sense to assume that $k$ is positive.
\end{framed}


Unfortunately, Major League Baseball does not seem to publish any values for $k$ (which would vary according to a variety of factors). They do however publish a variety of other pitch statistics. Some of them are as follows:
\begin{itemize}
\item {\it Velocity (VELO):} As stated above, this is the ball's maximum velocity achieved at any point between the pitcher's hand and home plate. 
\item {\it Extension (EXT):} This is how far in front of the pitcher's mound the ball is released.
\end{itemize}

In order to determine $k$ one needs measurements of time that the ball is in the air. The Diamond Kinetics Pitchtracker baseball \cite{Dataset} (a Bluetooth enabled baseball for pitch training) also measures the following:

\begin{itemize}
\item {\it Time to Plate:} This is the time it takes from the pitcher's initial movement until the ball gets to home plate, which is 60.5 feet from the pitcher's mound.
\item {\it Delivery Time:} This is the time from the pitcher's initial movement until the ball is released.
\end{itemize}



A collection of data from one pitcher in a study of the Pitchtracker ball at \cite{Dataset} is given by the following table:
\begin{center}
\begin{tabular}{|c | c | c | c|}
\hline Velocity (mph) & Time to Plate (s) & Delivery Time (s) & Extension (ft)\\
\hline 85 & 1.2 & .74 & 5.1 \\
 83.7 & 1.3 & .84 & 6.2 \\
 79.3 & 1.25 & .79 & 4.1 \\
 83.6 & 1.4 & .95 & 4.8 \\
 82.3 & 1.21 & .75 & 5.7 \\
 85.7 & 1.26 & .81 & 5 \\
 81.7 & 1.17 & .71 & 5.9\\
 86.7 & 1.24 & .79 & 4.5\\
 77.7 & 1.23 & .72 & 5.7\\
 72.7 & 1.25 & .74 & 3.2\\
 77.3 & 1.37 & .86 & 5.8\\
 76.9 & 1.25 & .74 & 5.8\\
 73 & 1.22 & .69 & 4.4\\
 77.1 & 1.24 & .78 & 6\\
 \hline 
\end{tabular}
\end{center}
Let's try to use this data to find $k$.

\begin{framed}
\noindent{\bf Task 2:} Solve the differential equation you derived in Task 1 analytically for velocity. Use this to solve for displacement, $s$, in feet from the pitcher's mound (consider the direction toward home plate to be the forward direction). Use this to attempt to find $k$ from any row of the above data  set. Explain where and why you get stuck.

\end{framed}

\begin{framed}
\noindent{\bf Task 3:} Use numerical/graphical methods of your choice to estimate $k$ as accurately you feel is possible from the above data set. Be sure to explain your process. Here are a few hints:
\begin{itemize}
\item For any row of the data set, you will need to begin with an initial guess for $k$ and narrow it down to match the data. It's good to know in advance that $k$ is quite small.
\item Some of the data is erroneous, so you need to keep an eye out for it.
\item To avoid frustration, be careful with your units(!).
\end{itemize}
\end{framed} 

\begin{framed}
\noindent{\bf Task 4:} The fastest pitch velocity ever recorded was by a 105.1 mph pitch by Aroldis Chapman on September 24, 2010. This velocity was measured 50.5 feet from home plate (in the PITCHf/x era) \cite{FastestPitches}. In 1974, Nolan Ryan had his fastball velocity clocked at 100.9 mph \cite{NolanRyan} (in the 9th inning of a game he started!). This measurement was taken 10 feet in front of home plate. Use your model to compare these pitches. Which pitcher really holds the title of being the hardest thrower?
\end{framed}








\teacher{
\section*{NOTES FOR TEACHERS}
\begin{enumerate}
\item The exponent of $v$ must be $2$ in order for the units on the right- and left-hand side of the equation to match.

\item Separation of variables and integration yields the position, in ft, from the pitcher's mound as
\begin{eqnarray*}
s(t) & =  & \frac{1}{k}\ln\left(kt + \frac{1}{v_0}\right) + EXT + \frac{1}{k}\ln(v_0)\\
 & =  & \ln\left(\sqrt[k]{kv_0t +1}\right) + EXT,
\end{eqnarray*}
where $EXT$ is the pitcher's extension and $v_0$ is the initial velocity in ft/s. This equation is not solvable in terms of elementary functions.


\item One strategy for finding $k$ is to graph with Desmos with $k$ as a slider and matching for the given data. For example \url{https://www.desmos.com/calculator/7c1iq8qff1} has the approximate value of $k$ for the first row in the given table. Students should proceed with finding an estimate of $k$ for each row of the table. Observe that rows 3, 4, 10, and 14 are erroneous data. From there, most students will take the average of the $k$ values they found. A more sophisticated method is to find $k$ using least squares (Excel sheet using the Solver add-in is available in the supporting docs); this yields a value of $k\approx .001149$.

Another method of approximating values of $k$ is experimentally via numerical methods. For instance, using the modeling software website {\it Insightmaker}, students can construct the model and run it for various values of $k$ to make the data match. A sample Insight for this scenario is available at \url{https://insightmaker.com/insight/1NnXT0rCZUBFVo5XIHxdVP/Fastball-Model}.

\item Using $k = .001149$ and an ``extension'' of 10 ft, we can see with Desmos that a pitch measured at 105.1 mph ($\approx 154.147$ fps) reaches a point 50.5 ft from the pitcher's mound in a time of $\approx .2689$ seconds. At that time, its velocity is $\approx 147.139$ fps, or about 100.3 mph. Similar results can be obtained with Insightmaker. Using a range of $k$-values can give a range of possible velocities for comparisons. The conclusion is that this model shows that these pitches were likely very close in velocity.  

\end{enumerate}    
}








\end{document}
